**Author’s response to reviews**

**Title:** The Resonance® metallic ureteral stent in the treatment of malignant ureteral obstruction: A prospective observational study

**Authors:**

Jun Miyazaki (jmiyazaki@iuhw.ac.jp)
Mizuki Onozawa (mizukionozawa@iuhw.ac.jp)
Satoshi Takahashi (stakahas@sapmed.ac.jp)
Yuka Maekawa (yuka_mae@gifu-u.ac.jp)
Mitsuru Yasuda (super7@gifu-u.ac.jp)
Koichiro Wada (gmd17055@s.okadai.jp)
Yuji Maeda (y-maeda@mattohp.com)
Takuro Masaki (t-masaki@harasanshin.or.jp)
Akito Yamaguchi (yamaguchi@harasanshin.or.jp)
Masahiko Suzuki (Masahiko12o11u08@aol.com)
Yasuyuki Sakai (yasakai@cancer-c.pref.saitama.jp)
Tomokazu Kimura (tomokazu.kim@md.tsukuba.ac.jp)
Manabu Takai (manabu11gaku@yahoo.co.jp)
Kensaku Seike (k.seike-04.6.26@leaf.ocn.ne.jp)
Takahiko Hashimoto (th-uro@hyo-med.ac.jp)
Shingo Yamamoto (shingoy@hyo-med.ac.jp)

**Version:** 4  **Date:** 11 Dec 2019

**Author’s response to reviews:**

1. In the Ethics approval and consent to participate subsection of your Declarations please include the full unabbreviated names of the ethics committees at participating institutions which approved your study protocol. If this list is too long or unwieldy for this subsection you may include the list as a figure, table, or supplementary file.

We add the list of ethics committees in the Declarations section.
2. If two authors have the same initials, please distinguish between them by designating them with numbers 1 and 2. For example: YM1 and YM2. YM1 would correspond to the author furthest up on the author list.

It was corrected as the editor pointed out.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

We uploaded a single, final, clean version.